Author’s response to reviews

Title: Determinants of Severe Acute Malnutrition Among Under Five Children In rural Enebsie Sarmidr District, East Gojjam Zone, North West Ethiopia, 2016

Authors:
Abate Awoke (abateawoke@yahoo.com)
Mulatu Ayana (mulatuayana@gmail.com)
Tenaw Gualu (tenawgualu@yahoo.com)

Version: 6 Date: 05 Sep 2017

Author’s response to reviews:
The manuscript is reviewed as commented.